Author’s response to reviews

Title: Measurement Properties of Oral Health Assessments for Non-dental Healthcare Professionals In Older People– a systematic review-

Authors:

Babette Everaars (babette.everaars@hu.nl)
Linet Weening-Verbree (l.f.weening-verbree@pl.hanze.nl)
Katarina Jerković – Ćosić (katarina.jerkovic@hu.nl)
Linda Schoonmade (l.j.schoonmade@vu.nl)
Nienke Bleijenberg (nienke.bleijenberg@hu.nl)
Niek de Wit (N.J.deWit@umcutrecht.nl)
Geert van der Heijden (geert.vander.heijden@acta.nl)

Version: 1 Date: 02 May 2019

Author’s response to reviews:

Rebuttal: BGTC-D-18-00306

Different approaches for assessing oral health of older people by non-dental healthcare professionals -A systematic Review Evaluating Their Measurement Properties- Babette Everaars, MSc.; Linet F. Weening-Verbree, MSc.; Katarina Jerković – Ćosić, PhD.; Linda Schoonmade, MA.; Nienke Bleijenberg, RN, PhD.; Niek J. de Wit, MD, PhD.; Geert J.M.G. van der Heijden, PhD.
BMC Geriatrics

Dear Tovah Honor Aronin and reviewers,

We would like to thank you very much for the insightful comments you have provided in our manuscript for further improvement of our manuscript.
In the attached file (Rebuttal), we give a reply per comment. If we made changes in the manuscript we refer to the corresponding page and line numbers and highlighted the changes in the text with green (reviewer 1) and yellow (reviewer 2).

We look forward to hear from you soon. We would be glad to respond to any further questions and comment if necessary.

On behalf of all co-authors.,

Yours sincerely,
Babette Everaars